Author's response to reviews

Title: Computerized acoustic assessment of treatment efficacy in RSV Bronchiolitis: Nebulized Epinephrine Versus Albuterol- a double-blind study.

Authors:

Raphael Beck (r.beck@rambam.health.gov.il)
Nael Elias (n.elias@rambam.health.gov.il)
Shay Shoval (shoval@hotmail.com)
Naveh Tov (mdnave@tx.technion.ac.il)
Gil Talmon (g.talmon@rambam.health.gov.il)
Simon Godfrey (sgodfrey@netvision.net.il)
Lea Bentur (l.bentur@rambam.health.gov.il)

Version: 4 Date: 27 May 2007

Author's response to reviews: see over
Point-by-point response

Original manuscript title: Computerized acoustic assessment of treatment efficacy in RSV bronchiolitis: nebulized epinephrine versus albuterol – a double-blind study.

Authors: R. Beck, MD, N. Elias, MD, S. Shoval, MD, N. Tov, MD, G. Talmon, MD, S. Godfrey, MD, PhD, L. Bentur, MD.

Reviewer 4 (R. Davis)

Minor essential revisions:

1. The p value column was removed. The sentence was changed to the suggested wording.

2. Due to the fact that there was no clear separation between the higher scores and lower scores, and the small number of patients that would end up in each group, we decided to eliminate this sentence altogether, as we feel that statistical analysis here would not be appropriate.

3. The p values have been added.

Discretionary revisions:

1. The capitalization was removed.
Formatting changes

1. Running head and keywords have been removed
2. The abstract in the submission has been replaced
3. The underlining has been removed
4. The text at the bottom left of the pages has been removed
5. "Conclusion" has been changed to "Conclusions"
6. "Authors contributions" has been changed to "Authors' contributions"
7. The figures were further cropped. They are submitted in MS Word format.

We thank the reviewers for their effort and their many helpful comments.